Author's response to reviews

Title: Measuring patient safety culture in Taiwan using the Hospital Survey on Patient Safety Culture (HSOPSC)

Authors:

I-Chi Chen (FL001@mail.oit.edu.tw)
Hung-Hui Li (hhlee@saturn.yzu.edu.tw)

Version: 8 Date: 3 June 2010

Author's response to reviews: see over
June 3, 2010

Editorial Board,
BMC Health Services Research

Dear editors,

We would like to thank you for accepting our paper, “Measuring patient safety culture in Taiwan using the Hospital Survey on Patient Safety Culture (HSOPSC)” (MS: 5727961023123037).

We have checked the manuscript and made two minor changes on page 1 (title page):
1. Change “Graduate School of Management” to “College of Management”
2. Change “FL001@mail.oit.edu.tw” to “s949407@mail.yzu.edu.tw”

Thank you.

Sincerely yours,

I-Chi Chen
Lecturer
Department of Health Care Administration,
Oriental of Technology,
58, Sec 2, Sihchuan Road, Pan Chiao City, Taipei County, Taiwan, R.O.C.
Tel: +886-2-77380145 ext 6223
Fax: +886-2-77380898
Email: FL001@mail.oit.edu.tw